Author's response to reviews

Title: Attachment and coping in psychosis in relation to spiritual figures
Attachment and religion in psychosis Philippe Huguelet1, M.D. Sylvia Mohr1, Ph.D., Isabelle Rieben2, Ph.D., M.A., Roland Hasler1, Ph.D., Nader Perroud1, M.D., Pierre-Yves Brandt2, Ph.D.

Authors:

Philippe Huguelet (Philippe.Huguelet@hcuge.ch)
Sylvia Mohr (Sylvia.Mohr@hcuge.ch)
Isabelle Rieben (isabelle.rieben@hcuge.ch)
Roland Hasler (roland.hasler@hcuge.ch)
Nader Perroud (nader.perroud@hcuge.ch)
Pierre-Yves Brandt (pierre-yves.brandt@unige.ch)

Version: 3 Date: 19 August 2015

Author's response to reviews: see over
Dear Editor,

Thank you very much for your 2nd reply to our submission of the above-mentioned manuscript and for the reviewer’s comments. Please find enclosed the revised version of our paper and our responses to the 2nd reviewer’s comments indicating how we have revised the manuscript.

**Reviewer's report:**
This revised version of 'Attachment and Coping in Psychosis' is improved and addresses major concerns outlined in my previous report. The paper reads much better, especially the discussion, and the rationale for the study is clearer.

There continues to be some minor grammatical errors that need to be addressed:

Minor Essential Revisions:
1. **Review sentence line 397:** "Data from several studies using auto-questionnaires to assess attachment dimensions lead for a good evidence between avoidant attachment and positive and negative symptoms"
   May instead read: "... assess attachment dimensions evidence a link between" or "... provide evidence for a link between..."
   Done

2. **Review sentence line 428**, starting with "Indeed...". Brackets are opened but not closed. "For example" is stated twice in sentence. May be better to break up into two sentences.
   Done

3. **Review sentence line 446**, should read "attachment was associated with a number of..."
   Done

We thank the reviewer for his/her careful revision and we hope that this revised version will make a valuable contribution to “BioMed Central”.

Yours sincerely,

On behalf of the authors,

Pr. Ph. Huguelet, MD